UC San Diego

UC San Diego Previously Published Works

Title

Benefits and risks of low-dose glucocorticoid treatment in the patient with rheumatoid arthritis.

Permalink

https://escholarship.org/uc/item/93t4s03b

Journal

Rheumatology, 53(10)

Authors

Kavanaugh, Arthur Wells, Alvin

Publication Date

2014-10-01

DOI

10.1093/rheumatology/keu135

Peer reviewed

Review

RHEUMATOLOGY

Benefits and risks of low-dose glucocorticoid treatment in the patient with rheumatoid arthritis

Arthur Kavanaugh¹ and Alvin F. Wells²

Abstract

Glucocorticosteroids (GCs) have been employed extensively for the treatment of rheumatoid arthritis (RA) and other autoimmune and systemic inflammatory disorders. Their use is supported by extensive literature and their utility is reflected in their incorporation into current treatment guidelines for RA and other conditions. Nevertheless, there is still some concern regarding the long-term use of GCs because of their potential for clinically important adverse events, particularly with an extended duration of treatment and the use of high doses. This article systematically reviews the efficacy for radiological and clinical outcomes for low-dose GCs (defined as ≤10 mg/day prednisone equivalent) in the treatment of RA. Results reviewed indicated that low-dose GCs, usually administered in combination with synthetic DMARDs, most often MTX, significantly improve structural outcomes and decrease symptom severity in patients with RA. Safety data indicate that GC-associated adverse events are dose related, but still occur in patients receiving low doses of these agents. Concerns about side effects associated with GCs have prompted the development of new strategies aimed at improving safety without compromising efficacy. These include altering the structure of existing GCs and the development of delayed-release GC formulations so that drug delivery is timed to match greatest symptom severity. Optimal use of low-dose GCs has the potential to improve long-term outcomes for patients with RA.

Key words: rheumatoid arthritis, glucocorticoids, prednisone, disease modifying, treatment strategies, benefit-risk.

Introduction

The role of glucocorticoids in the treatment of RA

Glucocorticosteroids (GCs) have a long history of good efficacy and safety in the treatment of RA. This has resulted in their inclusion in guidelines for the management of this disease. For example, the European League Against Rheumatism (EULAR) guidelines recommend that GCs be added at low to moderately high doses to synthetic DMARD monotherapy (or combinations of synthetic DMARDs) since they have been shown to provide benefit as an initial short-term treatment. However, it is also generally recommended that GCs should be tapered as rapidly as clinically feasible [1]. The Canadian Rheumatology Association treatment recommendations state that GCs (oral, intramuscular or intra-articular) can be added to DMARD therapy as part of the initial treatment strategy for patients with RA, and may be an option for managing flares as bridge therapy while waiting for a DMARD to take effect, or for symptom control if no other options exist [2].

While the use of GCs in patients with RA is supported by clinical trial results, there is still some concern surrounding their use because of potential associations with clinically important adverse events, particularly when they are administered at high doses and/or for long duration [3-5]. While caution is certainly warranted in the use of GCs in patients with RA, results from a large number of studies have indicated that combination of low doses of these agents with DMARDs may have significant benefit with respect to joint preservation and also acceptable safety. In addition, new GC molecules are being developed in an effort to improve their efficacy and

The aim of this article was to systematically review the efficacy (radiological and clinical outcomes) and safety of low-dose GCs when used as part of treatment regimens for patients with RA. The analysis included studies in

Correspondence to: Arthur Kavanaugh, Department of Medicine, University of California, San Diego, 9500 Gilman Drive, La Jolla, CA 92093-0943, USA. E-mail: akavanaugh@ucsd.edu

¹Department of Medicine, University of California, San Diego, La Jolla, CA and ²Department of Rheumatology, Duke University Medical Center, Durham, NC, USA.

Submitted 28 February 2013; revised version accepted 3 February 2014.

which GCs were administered at low doses ($\leq 10 \, \text{mg/day}$ prednisone equivalent). In some trials, GCs were delivered at higher doses for a short interval prior to tapering to a dose $\leq 10 \, \text{mg/day}$.

Literature review

PubMed was searched using the following guery, rheumatoid AND (hydrocortisone OR methylprednisolone OR budesonide OR betamethasone OR dexamethasone OR prednisone OR prednisolone OR corticosteroid OR glucocorticoid). The results were limited to randomized controlled trials with the appearance of the words in the title or abstract. This search resulted in retrieval of 218 studies. The abstract or, when necessary, full text of each study was then reviewed for relevance for this article using the following criteria: (i) treatment of adult patients with RA, (ii) use of oral GC doses ≤10 mg/day, (iii) treatment arms and data presentation that permitted assessment of the effects of GCs on treatment outcomes (e.g. studies in which GCs were permitted along with other treatments in multiple study arms; studies in which results were not presented separately for patients receiving and not receiving GCs were excluded). Secondary and redundant publications of results from a given study were also excluded. The reference sections of the papers included in the review were also evaluated for additional citations not recovered by the PubMed search. The search resulted in retrieval of 55 papers that met the above-listed criteria. This total included non-redundant reports at different time points for several long-term studies, all of which are included in the analysis.

Effects of GCs on structural outcomes in patients with RA

Fifteen of the studies retrieved provided information about radiological outcomes for patients receiving low-dose GCs (Table 1). Analysis of the results from these 15 studies supports the view that administration of GCs in combination with synthetic DMARDs, most often MTX, significantly improves structural outcomes in patients with RA. Results from all 15 clinical trials with results for this endpoint are summarized in Table 1. One of the more convincing demonstrations of the efficacy of adding GCs to conventional treatment for RA is the Combination Therapy of RA (COBRA) trial. In this study, 155 patients with early RA received SSZ monotherapy (SSZ group) or a combination of step-down prednisolone, MTX and SSZ (COBRA group). After 11 years of follow-up, less progression of joint damage was observed in the COBRA group [11]. However, it should also be noted that the incidence of treatment for hypertension, diabetes and cataracts were all significantly higher in the COBRA group vs the SSZ group. Conversely, hypercholesterolaemia, cancer, infection and mortality all showed trends favouring initial high-dose GC treatment in COBRA [11]. A second longterm trial, the BeSt study, compared four treatment strategies: sequential substitution DMARD monotherapy, step-up add-on DMARD combination therapy, initial

combination therapy that included prednisone (60 mg/day initial dose tapered to 7.5 mg/day over 7 weeks; similar to that in COBRA) and initial combination therapy with a DMARD plus the TNF inhibitor infliximab.

Results from the BeSt study showed that combination therapy that included initial high-dose prednisone slowed joint erosion vs sequential monotherapy or step-up combination treatment that did not include a high-dose GC [13-15]. After 5 years, the initial benefit of combination therapy was maintained, with significantly less joint damage than in patients who did not receive a GC [16]. The Better Anti-rheumatic Farmacotherapy (BARFOT) study showed that the addition of prednisolone 7.5 mg/ day to initial DMARD therapy retarded the progression of erosions after 2 years in patients with early RA and provided a higher remission rate than DMARD therapy alone. Remission achieved after 2 years was associated with less radiographic damage after 4 years, which when analysed according to initial treatment group, was statistically significant only for patients receiving early GC therapy [19, 20]. Results from the COBRA, BeSt and BARFOT studies are consistent with those from a large number of additional studies that assessed radiographic damage (e.g. joint erosion) in patients with RA who received GCs for shorter periods (Table 1).

Not all studies retrieved demonstrated a significant benefit of GC treatment on structural outcomes (Table 1). A trial that included 167 patients with RA who were followed for 2 years indicated no radiological or clinical benefit of adding 7 mg/day prednisolone to SSZ [21]. A 3-year study compared radiographic progression rates in paired hand radiographs in patients taking NSAIDs alone or NSAIDs plus ≤ 5 mg/day prednisolone. Study results indicated that for a subset of 252 patients with an RA duration of 12–24 months, changes from baseline radiographic scores were not different in those taking or not taking prednisone ($P\!=\!0.994$) [23].

Effects of GCs on clinical outcomes: efficacy

The majority of the 55 studies reviewed indicated that treatment with low-dose GC alone *vs* placebo or a combination of low-dose GCs with DMARDs *vs* DMARDs only significantly improved outcomes for measures that included ACR responses [21, 24], HAQ scores [14, 25], tender and swollen joint counts [24] and DAS [26] (see supplementary Table S1, available at *Rheumatology* Online).

An example of the effects of low-dose GC monotherapy on clinical endpoints is provided by a 12-week, double-blind, randomized controlled trial in which 143 patients with active RA received budesonide 3 or $9\,\text{mg/day}$, prednisolone 7.5 mg/day or placebo. Treatment with either $9\,\text{mg/day}$ budesonide or prednisolone significantly decreased tender and swollen joint counts vs placebo (P < 0.05) and ACR20 responses were achieved by 22% and 42% of patients who received budesonide 3 or $9\,\text{mg/day}$ and 56% of those on prednisolone vs 25%

TABLE 1 Radiological outcomes for studies in which low-dose GCs were included in the treatment of RA patients

Reference	Patients	Duration, design and treatment	Evaluations/endpoint	Outcome
van Everdingen et al. [6]	81 patients with early active RA who had not been treated with DMARDs	2-year randomized, double-blind, placebo- controlled clinical trial Patients received 10 mg prednisone or placebo NSAIDs were allowed in both groups and SSZ could be prescribed for rescue	Radiological studies were performed every 6 months	After month 6, radiological scores showed significantly less progression in the prednisone group than in the placebo group
Hickling <i>et al.</i> [7]	128 patients with early active RA	2-year, randomized, double-blind, placebo-controlled trial 7.5 mg prednisolone daily in addition to routine medication over 2 years in 128 patients with early RA	Annual radiological evaluation of the hand with Larsen scoring	Prednisolone treatment resulted in significantly fewer erosive changes vs placebo Withdrawal of prednisone during the third year of the study resulted in disease progression similar to that observed in the control group
Rau <i>et al.</i> [8]	196 patients with active RA	2 year, double-blind, randomized, multi- centre study Patients receive 5 mg prednisolone or placebo Patients also received DMARD treatment with either gold sodium thiomalate or MTX	Radiographs were taken at 6, 12 and 24 months and evaluated using the Ratingen score and van der Heijde's modification of Sharp's method	After 24 months the total score had increased by 2.6% of the maximum score in the placebo group and by 1.1% in the prednisolone group
Boers <i>et al.</i> [9], Landewé <i>et al.</i> [10], van Tuyl <i>et al.</i> [11]	155 patients with early RA	1-year, multicentre, double-blind, randomized trial Patients received a combination of SSZ (2g/day), MTX (7.5 mg/week) and prednisolone (initially 60 mg/day, tapered in 6 weekly steps to 7.5 mg/day) or SSZ alone Prednisolone and MTX were tapered and stopped after 28 and 40 weeks, respectively.	Sharp-van der Heijde radiographic damage score in hands and feet	After 26 weeks the median radiographic damage scores increased by 1 in the combination treatment groups vs 4 for SSZ. The respective values at week 80 were 4 vs 12 During the 4- to 5-year follow-up period, the Sharp progression rate was 8.6 points/year in the SSZ group and 5.6 in the combination treatment group The 11-year follow-up also indicated less joint damage in the combination treatment group
Malysheva <i>et al.</i> [12]	154 patients with RA	Retrospective analysis of patients followed for 2-62 months Various treatments, including GCs (7.5mg prednisone equivalent), MTX, SSZ, HCQ, and AZA	Routine radiological evaluation	A significant reduction was observed in the frequency of erosive RA in patients with GC co-medication vs those without low-dose GCs

7	7
ă	í
=	5
2.	Ξ
+	2
7	
\sim	í
_	•
$\overline{}$	-
ш	ı
	j
0	ฉี

Reference	Patients	Duration, design and treatment	Evaluations/endpoint	Outcome
Goekoop- Ruiternan <i>et al.</i> [13–15], Klarenbeek <i>et al.</i> [16], van der Kooij <i>et al.</i> [17]	508 patients with early RA	Multicentre randomized trial with four treatment groups: (i) sequential monotherapy, (ii) step-up combination therapy, (iii) initial combination therapy with tapered high-dose prednisone and (iv) initial combination therapy with infliximab	Radiographic joint damage according to the modified Sharp-van der Heijde score	At 1 year, median increases in the total Sharp-van der Heijde radiographic joint score were 2.0, 2.5, 1.0 and 0.5 in groups 1-4, respectively After 5 years, initial combination therapy resulted in significantly less joint damage progression
Svensson et al. [18], Svensson et al. [19], Hafstrom et al. [20]	187 patients with early RA	Long-term multicentre observational study of patients with early RA in southern Sweden Within this study, 187 patients were randomized to open label: (i) initial treatment for 1 month with 7.5 mg prednisolone (for 28 patients, MTX was added) or (ii) SSZ or AUR	Radiographic evaluation at 2 years with Larsen scoring	Addition of prednisolone to initial DMARD therapy retarded the progression of erosions after 2 years in patients with early RA and provided a higher remission rate than DMARD therapy alone Remission achieved after 2 years was associated with less radiographic damage still present after 4 years, which when analysed according to initial treatment group was statistically significant only for patients receiving early prednisolone therapy
Capell <i>et al.</i> [21]	167 patients with RA for <3 years	2-year, randomized, double-blind, pla- cebo-controlled trial Patients were started on SSZ and rando- mized to prednisolone 7 mg/day or	Radiological damage as assessed by the modified Sharp method	There were no significant between- group differences in radiological scores or clinical and laboratory measures at 2 years
Kirwan e <i>t al.</i> [22]	128 patients with active RA for <2 years	2-year, randomized, double-blind, pla- cebo-controlled trial Patients were randomized to 7.5 mg of prednisolone or placebo plus any other prescribed medication	Progression of damage as seen on radiographs of the hand after 1 and 2 years, as measured by the Larsen index, and the appearance of erosions in hands that had no erosions at baseline	After 2 years, Larsen scores increased by a mean of 0.72 U in the prednisolone group and by 5.37 U in the placebo group (P = 0.004). Of the 212 hands of these patients, 69.3% had no erosions at the start of the study. At 2 years, 22.1% of those in the prednisolone group and 45.6% of those in the placebo group had erosions (P = 0.007)
Paulus <i>et al.</i> [23]	824 patients with RA	3-year prospective, randomized clinical trial comparing the NSAIDs etodolac and ibuprofen DMARDs were not permitted Prednisone ≤5 mg/day was continued by 197 patients (mean dose = 4.37 mg/day) who had started prednisone therapy ≥6 months before study entry, but new prednisone starts were not allowed	Rate of increase in erosion scores	For the subgroup of 252 patients with RA duration of 12-24 months, changes from baseline in radiographic scores (mean monthly erosion) were not different in those taking or not taking prednisone (P=0.994)

AUR: auranofin; GCs: glucocorticosteroids.

for placebo. Results for both 9 mg/day budesonide (P < 0.001) and prednisolone (P = 0.02) were significantly different from placebo [24]. The benefit of adding lowdose GC to MTX-based treatment was demonstrated by results from the Computer-Assisted Management in Early Rheumatoid Arthritis (CAMERA) II trial. This study included 236 patients with early RA (<1 year) who received either MTX or MTX plus 10 mg prednisone and were followed for 2 years. Results for the 28-joint DAS (DAS28), pain and HAQ scores decreased more rapidly in patients who received MTX plus prednisone vs those on MTX alone, but differences between treatment groups declined over the course of follow-up. Longitudinal regression analysis indicated lower significantly better improvement for all disease activity variables with MTX plus prednisone vs MTX alone (P < 0.001). At the end of the 2-year followup period, 65% of patients receiving MTX plus prednisone and 61% on MTX only achieved a 20% improvement in ACR criteria (ACR20; P = 0.56); the respective values for ACR50 were 53% and 42% (P = 0.091) and those for ACR70 were 38% and 19% (P=0.002) [27].

There is considerable ongoing debate about the optimal means of defining remission in RA, and various measures have been suggested [28]. Nevertheless, remission has come to be considered the primary therapeutic goal of RA treatment [1, 2]. Results from several studies, using varying definitions, have shown that the addition of GC to the treatment regimen for a patient with RA may increase the probability of achieving disease remission [19, 27, 29-32] (see supplementary Table S1, available at Rheumatology Online). Results from a study of 105 patients with early active RA indicated that the addition of low-dose prednisone to treatment significantly increased the probability of achieving and sustaining remission over 2 years of follow-up [31]. A randomized trial of 220 patients with active RA <2 years from symptom onset indicated that addition of 6.25 mg/day prednisone to MTX resulted in significant increases in both clinical (DAS28 <2.6) and ultrasonographic remission rates vs MTX plus placebo over 12 months of follow-up [30]. Results from the CAMERA II trial, a 2-year prospective study that included 236 patients with early RA (<1 year), indicated that the combination of prednisone and MTX was superior to MTX alone in slowing erosive joint damage, reducing disease activity and physical disability and achieving sustained remission [27]. Results from the Finnish RA combination therapy trial showed that inclusion of prednisone with other conventional DMARDs in 199 patients with RA significantly increased the probability of achieving the modified minimum disease activity (MDA) and strict ACR criteria for remission over 11 years of follow-up [32].

It is important to note that not all of the studies reviewed indicated sustained clinical benefit of low-dose GC in patients with RA. A study of 81 patients with early RA (<1 year duration) indicated that $10\,\mathrm{mg/day}$ prednisone was significantly superior to placebo for decreasing pain at 3 months (P=0.003) and improving general well-being at 3 and 6 months (P=0.003) and P=0.04, respectively), but not thereafter over a 2-year follow-up period [33].

Effects of GCs on clinical outcomes: safety

The adverse events associated with GC treatment in patients with RA have been enumerated in many publications over the years. Commonly reported adverse events include weight gain, elevated blood pressure (BP), glucose intolerance and increased risk for diabetes and its complications, development of cataracts and increased incidence of glaucoma, increased susceptibility to infections, hyperlipidaemia, gastrointestinal adverse events (e.g. ulcers, lower gastrointestinal bleeding, perforations), development of osteoporosis and insufficiency fractures, osteonecrosis and changes in physical appearance (e.g. Cushingoid appearance, hirsutism, abdominal striae) [34-37]. GC-associated adverse events are significantly associated with the daily dose and treatment duration [38]. An important guestion related to long-term use of low-dose GCs in patients with RA is the extent to which risks for GC-associated adverse events might be attenuated at lower doses of these agents.

The results from the clinical trials included in this systematic review indicate that at least some of the adverse events observed with high-dose GCs also occur with lower doses of these drugs. The studies reviewed indicated that treatment with low-dose GCs were associated with weight gain [21, 39], hyperglycaemia [30] and diabetes [11, 19], increased blood pressure [21, 40, 41] and hypertension [11, 42], decreased BMD [42–45], increased risk for fractures [46], cognitive dysfunction [47], increased risk of infection [48] and cataracts [11].

Discussion

The results summarized in the preceding sections, Table 1 and supplementary Table S1 (available at Rheumatology Online) support the conclusion that the addition of lowdose GC to treatment with conventional DMARDs significantly improves both clinical and radiological outcomes in patients with RA, but that there are still safety concerns associated with the long-term use of these agents. The results of this review concerning radiological outcomes with GCs are consistent with those from two previously published meta-analyses. Results from an analysis of 15 trials that included 1414 patients (mean cumulative GC dose of 2300 mg prednisone equivalent/year) indicated that GCs given in addition to standard therapy can substantially reduce the rate of erosions in RA. For all the studies, the standardized mean difference in progression was 0.40 in favour of GCs (95% CI 0.27, 0.54). In studies lasting at least 2 years, the standardized mean difference in progression in favour of GCs at the end of the study period was 0.42 (95% CI 0.30, 0.55). All studies except one showed a numerical treatment effect in favour of GCs [49]. A more recent meta-analysis carried out to inform EULAR recommendations for the treatment of RA indicated that the addition of GCs to either standard synthetic DMARD monotherapy or combinations of synthetic DMARDs yielded clinical benefits and inhibition of

TABLE 2 Adverse events with low-dose GCs

	No CCo in post	Patients with GC intake for >6 months		
	No GCs in past 12 months, %	<5 mg/day, %	5-7.5 mg/day, %	> 7.5 mg/day, %
Cushingoid phenotype	2.7	4.3	15.8	24.6
Ecchymosis	6.8	17.4	23.5	24.6
Leg oedema	9.5	11.6	20.2	26.2
Mycosis	4.5	5.8	6.6	8.2
Parchment-like skin	3.2	10.1	15.8	21.3
Shortness of breath	9.5	10.1	12.6	16.4
Sleep disturbance	20.7	33.3	37.2	44.3
Eye cataract	2.7	10.1	7.7	8.2
Epistaxis	1.4	1.4	6.6	4.9
Weight gain	9.5	8.7	22.4	21.3
Depression, listlessness	12.6	10.1	13.7	19.7
Glaucoma	2.7	2.9	2.7	6.6
Increase in blood pressure	18.9	18.8	16.4	23.0

GCs: glucocorticosteroids. Modified from Huscher et al. [36] with permission from the copyright holder, BMJ Publishing Group Ltd.

radiographic progression that may extend over many years [50].

The benefits of low-dose GCs in patients with RA have led to multiple statements that these agents should be employed as part of the initial therapeutic regimen for patients with RA [51-54].

The safety of low-dose GCs was addressed in a combined analysis of results from four placebo-controlled trials in which RA patients received GC doses ≤10 mg/ day and were followed for 2 years. The results from this analysis indicated no excess cardiovascular events with low-dose GCs; no effect on BMD; a small rise in plasma glucose (from 5.1 to 5.9 mmol/l) in only one of the four studies; no increase in risk for cataracts, but increased risk for glaucoma; no increased risk for upper gastrointestinal ulcers and bleeds; no increased risk for infections and no increased risk for dermatological adverse events [35]. Analysis of 1066 patients included in the German Collaborative Centres database who received GCs for >6 months at doses of <5, 5-7.5 and >7.5 mg/day indicated that different adverse events with these drugs may have different thresholds (Table 2). The risk for some adverse events (e.g. ecchymosis, parchment-like skin, sleep disturbance and cataracts) appears to be increased with any GC dose, while others only increase with doses ≥5 mg (e.g. Cushingoid symptoms, leg oedema, weight gain) or >7.5 mg/day (e.g. shortness of breath, epistaxis, depression/listlessness, glaucoma) [36].

The concerns about side effects associated with GCs have prompted substantial efforts to improve safety profiles without compromising efficacy. Several alternative strategies have been tested in an effort to achieve this highly desirable goal.

GCs with altered structural characteristics aimed at decreasing the risk for adverse events have been evaluated in patients with RA. Deflazacort, an oxazoline

derivative of prednisolone with anti-inflammatory and immunosuppressive activity, first became available in 1969. The severity of steroid-induced osteoporosis and growth retardation due to deflazacort is less than that associated with other steroids [55]. Multiple randomized clinical trials appeared to indicate that deflazacort had less severe side effects than conventional GCs, but the dose to achieve and maintain equivalent anti-inflammatory efficacy was usually higher than predicted [56]. At these higher doses, the safety advantages of deflazacort were minimized [57]. It is also important to note that the bone-sparing effects of agents used to treat RA have generally been evaluated in small-scale, short-term studies. Well-designed long-term, large-scale trials are needed to address this issue.

Selective GC receptor agonists (SEGRAs) represent another approach to modification of the steroid molecule with the aim of improving safety. The classical genomic mechanism of GC action involves transrepression, which has been suggested to be primarily responsible for a large number of the desirable anti-inflammatory and immunomodulating effects of these drugs, and transactivation, which is suggested to be more closely associated with side effects as well as with some immunosuppressive activities [58]. This understanding has prompted the development of SEGRAs aimed at retaining the benefits of conventional GCs while decreasing their adverse effects [59]. To our knowledge, SEGRAs have not yet been evaluated in patients with RA.

So-called soft drug approaches have been used to design new drugs by attempting to alter molecules in order to impact their activity, as well as the most desired way for deactivation and detoxification. This approach has been used for GCs with the molecule loteprednol etabonate [60]. Loteprednol has been shown to be as effective as dexamethasone when injected directly into the

joint of rabbits with experimentally induced arthritis [61]. Agents in this class have not yet been evaluated in patients with RA.

A macromolecular pro-drug of dexamethasone, *N*-(2-hydroxypropyl) methacrylamide (HPMA) copolymer-dexamethasone conjugate, has been developed with the aim of selectively targeting inflamed tissues. Results from studies of animal models of inflammatory diseases, including LN, have indicated that this dexamethasone formulation provides sustained resolution of inflammation [62, 63]. Studies have not yet been reported in patients with RA.

New GC formulations have also been developed. Encapsulation of conventional GCs in liposomes has the potential to improve efficacy and decrease adverse effects by targeting delivery more directly to inflammatory sites and decreasing drug exposure to normal tissue. This approach has shown promise in animal models of arthritis, and a liposomal prednisolone preparation has been tested in a small number of patients with RA [64–66].

Efforts have also been directed towards optimizing the timing of low-dose GC delivery. RA symptoms have a circadian rhythm with the greatest severity in the early morning. A number of hypotheses have been put forward to explain this, including redistribution of interstitial fluids, circadian changes in synovial fluid composition and resultant oedema of the synovium and the peri-articular structures that interfere with joint biomechanics and nocturnal elevation in inflammatory mediators such as IL-6 that are not counterbalanced by a concomitant increase in cortisol [67, 68]. Increased production of melatonin at night may also contribute to the circadian pattern of clinical symptoms of RA [69]. Until recently, only two smallscale studies had evaluated the benefits of delivering GCs with regimens other than once-daily dosing at an unspecified time during the day. One small-scale trial compared once- and twice-daily GC administration (dose range 5-10 mg/day) in a small group of 20 RA patients. Qualitative evaluation indicated no efficacy differences between the two regimens [70]. A second 2-week randomized trial compared administration of indomethacin (100 mg) and prednisolone (5 mg) at night in 24 patients with RA. Both treatments significantly decreased morning stiffness and increased grip strength [71].

Delivery of a delayed-release prednisone formulation aimed at combatting the effects of the night-time increases in proinflammatory cytokines has been shown to be effective in RA. A delayed-release prednisone formulation has been developed that begins to release the drug ~4 h after ingestion. Taking this preparation at bedtime (approximately 10 P.M.) would result in the initiation of drug release at approximately 2 A.M., thereby theoretically more closely matching the increases in proinflammatory cytokines [3]. This preparation has been evaluated in two studies of patients with RA. In a 12-week, multicentre, randomized, double-blind trial, 288 patients with active RA were randomly assigned to either delayed-release prednisone or to immediate-release prednisone. The mean relative change in duration of morning stiffness of

the joints from baseline to the end of treatment was significantly higher with delayed-release prednisone than with immediate-release prednisone (-22.7% vs -0.4%, P = 0.045). In this study, delayed-release prednisone also decreased IL-6 concentrations by ~50% vs no change for immediate-release prednisone [72]. Another similar 12-week study included 350 patients with active RA who were randomized to receive delayed-release prednisone 5 mg or placebo. The primary endpoint was the percentage of patients achieving an ACR20. Delayed-release prednisone plus DMARD treatment produced higher ACR20 (48% vs 29%, P < 0.001) and ACR50 (22% vs 10%, P < 0.006) responses and a greater median relative reduction from baseline in morning stiffness compared with placebo (55% vs 35%, P < 0.002) [73]. A 9-month extension of these studies has been reported [67].

Safety results obtained to date with this delayed-release GC indicate that it did not appear to significantly suppress the hypothalamic-pituitary-adrenal axis [74]. Patients treated for ~9 months in extensions of the phase 3 trials reported a total of 51 serious adverse events. Bone fractures occurred in three patients and tendon rupture in one. The other serious adverse events included gynaecological problems, cardiovascular disease, respiratory disease, joint replacement surgery and synovectomy. Much more information about the side effects associated with long-term administration of this new GC formulation is needed to fully understand its safety profile.

An important issue that has not been addressed in the development of new GCs and formulations is resistance to these drugs, which has been suggested to occur in as many as 30% of patients with RA [75]. Mechanisms underlying GC resistance have been studied most extensively in the setting of chronic obstructive pulmonary disease and several possibilities have been supported. These include cytokine activation of mitogen-activated protein (MAP) kinase pathways, excessive activation of transcription factor activator protein 1, reduced histone deacetylase-2 (HDAC2) expression, increased levels of macrophage migration inhibitory factor and increased P-glycoprotein-mediated drug efflux [76]. Some these resistance mechanisms may be at least partially reversible. For example, it has been suggested that elevated HDAC2 expression can be reduced by administration of theophylline. In addition, several p38 MAP kinase inhibitors are currently in development that might impact steroid resistance [76].

Conclusions

With increases in the number and mechanisms of action of potent traditional and biologic DMARDs available for the treatment of RA, the goals of treatment have been raised. Thus there is still an unmet need to allow ever more patients to achieve the highest levels of disease control [77]. There is substantial evidence that low-dose GC treatment, when combined with conventional DMARDs, can significantly slow disease progression and increase the number of patients who achieve disease

remission. Therefore GCs remain a useful adjunctive treatment for RA and for other systemic inflammatory and autoimmune disorders. Physicians and other health care providers have had concerns about long-term GC treatment because of the well-known adverse events associated with high doses of these drugs. However, adverse events with GCs are dose related and longer-term use of low-dose GCs may still be a viable therapeutic option for some patients.

Multiple approaches have been undertaken to improve the benefit-risk profile for GC treatment in patients with RA, including the development of a delayed-release low-dose prednisone formulation that has shown some clinical benefit. Additional approaches are in earlier stages of development. Optimal use of low-dose GCs has the potential to improve long-term outcomes for patients with RA.

Rheumatology key messages

- Addition of low-dose glucocorticosteroids to synthetic DMARDs improves structural outcomes, decreases inflammatory markers and reduces symptom severity in RA patients.
- New approaches to glucocorticoid delivery may further improve RA outcomes.

Acknowledgements

The authors acknowledge Robert W. Rhoades, PhD and Valorie J. Thompson, BSN for editorial assistance.

Funding: This work was supported by Definitive Interventions in New Onset Rheumatic Arthritides (DINORA, Inc.), a 501(c)(3) corporation.

Disclosure statement: A.F.W. has worked as a consultant for Horizon Pharma USA. A.K. has declared no conflicts of interest.

Supplementary data

Supplementary data are available at *Rheumatology* Online.

References

- 1 Smolen JS, Landewé R, Breedveld FC et al. EULAR recommendations for the management of rheumatoid arthritis with synthetic and biological disease-modifying antirheumatic drugs. Ann Rheum Dis 2010;69:964-75.
- 2 Bykerk VP, Akhavan P, Hazlewood GS et al. Canadian Rheumatology Association recommendations for pharmacological management of rheumatoid arthritis with traditional and biologic disease-modifying antirheumatic drugs. J Rheumatol 2012;39:1559–82.
- 3 Spies CM, Cutolo M, Straub RH, Burmester GR, Buttgereit F. Prednisone chronotherapy. Clin Exp Rheumatol 2011;29(Suppl 68):S42-5.
- 4 van Tuyl LH, Plass AM, Lems WF et al. Why are Dutch rheumatologists reluctant to use the COBRA treatment

- strategy in early rheumatoid arthritis? Ann Rheum Dis 2007:66:974-6.
- 5 Weisman MH. Corticosteroids in the treatment of rheumatologic diseases. Curr Opin Rheumatol 1995;7:183–90.
- 6 van Everdingen AA, Jacobs JW, Siewertsz Van Reesema DR, Bijlsma JW. Low-dose prednisone therapy for patients with early active rheumatoid arthritis: clinical efficacy, disease-modifying properties, and side effects: a randomized, double-blind, placebo-controlled clinical trial. Ann Intern Med 2002;136:1-12.
- 7 Hickling P, Jacoby RK, Kirwan JR. Joint destruction after glucocorticoids are withdrawn in early rheumatoid arthritis. Arthritis and Rheumatism Council Low Dose Glucocorticoid Study Group. Br J Rheumatol 1998;37:930-6.
- 8 Rau R, Wassenberg S, Zeidler H, Study Group LDPT. Low dose prednisolone therapy (LDPT) retards radiographically detectable destruction in early rheumatoid arthritis—preliminary results of a multicenter, randomized, parallel, double blind study. Z Rheumatol 2000;59(Suppl 2):II/90-6.
- 9 Boers M, Verhoeven AC, Markusse HM et al. Randomised comparison of combined step-down prednisolone, methotrexate and sulphasalazine with sulphasalazine alone in early rheumatoid arthritis. Lancet 1997;350:309–18.
- 10 Landewé RB, Boers M, Verhoeven AC et al. COBRA combination therapy in patients with early rheumatoid arthritis: long-term structural benefits of a brief intervention. Arthritis Rheum 2002;46:347–56.
- 11 van Tuyl LHD, Boers M, Lems WF *et al.* Survival, comorbidities and joint damage 11 years after the COBRA combination therapy trial in early rheumatoid arthritis. Ann Rheum Dis 2010;69:807–12.
- 12 Malysheva OA, Wahle M, Wagner U et al. Low-dose prednisolone in rheumatoid arthritis: adverse effects of various disease modifying antirheumatic drugs. J Rheumatol 2008;35:979–85.
- 13 Goekoop-Ruiterman YP, de Vries-Bouwstra JK, Allaart CF et al. Clinical and radiographic outcomes of four different treatment strategies in patients with early rheumatoid arthritis (the BeSt study): a randomized, controlled trial. Arthritis Rheum 2005;52:3381-90.
- 14 Goekoop-Ruiterman YP, de Vries-Bouwstra JK, Allaart CF et al. Comparison of treatment strategies in early rheumatoid arthritis: a randomized trial. Ann Intern Med 2007;146:406-15.
- 15 Goekoop-Ruiterman YP, de Vries-Bouwstra JK, Allaart CF et al. Clinical and radiographic outcomes of four different treatment strategies in patients with early rheumatoid arthritis (the BeSt study): a randomized, controlled trial. Arthritis Rheum 2008;58(Suppl):S126-35.
- 16 Klarenbeek NB, Guler-Yuksel M, van der Kooij SM et al. The impact of four dynamic, goal-steered treatment strategies on the 5-year outcomes of rheumatoid arthritis patients in the BeSt study. Ann Rheum Dis 2011;70:1039–46.
- 17 van der Kooij SM, Goekoop-Ruiterman YP, de Vries Bouwstra JK et al. Drug-free remission, functioning and radiographic damage after 4 years of response-driven treatment in patients with recent-onset rheumatoid arthritis. Ann Rheum Dis 2009;68:914–21.
- 18 Svensson B, Schaufelberger C, Teleman A, Theander J. Remission and response to early treatment of RA assessed by the disease activity score. BARFOT

- study group. Better Anti-rheumatic Farmacotherapy. Rheumatology 2000;39:1031-1036.
- 19 Svensson B, Boonen A, Albertsson K *et al.* Low-dose prednisolone in addition to the initial disease-modifying antirheumatic drug in patients with early active rheumatoid arthritis reduces joint destruction and increases the remission rate: a two-year randomized trial. Arthritis Rheum 2005;52:3360-70.
- 20 Hafström I, Albertsson K, Boonen A et al. Remission achieved after 2 years treatment with low-dose prednisolone in addition to disease-modifying anti-rheumatic drugs in early rheumatoid arthritis is associated with reduced joint destruction still present after 4 years: an open 2-year continuation study. Ann Rheum Dis 2009;68: 508–13
- 21 Capell HA, Madhok R, Hunter JA et al. Lack of radiological and clinical benefit over two years of low dose prednisolone for rheumatoid arthritis: results of a randomized controlled trial. Ann Rheum Dis 2004;63:797–803.
- 22 Kirwan JR. Arthritis and Rheumatism Council Low-dose Glucocorticoid Study Group. The effect of glucocorticoids on joint destruction in rheumatoid arthritis. N Engl J Med 1995;333:142-6.
- 23 Paulus HE, Di Primeo D, Sanda M et al. Progression of radiographic joint erosion during low dose corticosteroid treatment of rheumatoid arthritis. J Rheumatol 2000;27: 1632-7.
- 24 Kirwan JR, Hällgren R, Mielants H et al. A randomised placebo controlled 12 week trial of budesonide and prednisolone in rheumatoid arthritis. Ann Rheum Dis 2004; 63:688–95.
- 25 Hansen M, Podenphant J, Florescu A et al. A randomised trial of differentiated prednisolone treatment in active rheumatoid arthritis. Clinical benefits and skeletal side effects. Ann Rheum Dis 1999;58:713–8.
- 26 van Gestel AM, Laan RF, Haagsma CJ, van de Putte LB, van Riel PL. Oral steroids as bridge therapy in rheumatoid arthritis patients starting with parenteral gold. A randomized double-blind placebo-controlled trial. Br J Rheumatol 1995;34:347–51.
- 27 Bakker MF, Jacobs JW, Welsing PM et al. Low-dose prednisone inclusion in a methotrexate-based, tight control strategy for early rheumatoid arthritis: a randomized trial. Ann Intern Med 2012;156:329-39.
- 28 Iking-Konert C, Aringer M, Wollenhaupt J et al. Performance of the new 2011 ACR/EULAR remission criteria with tocilizumab using the phase IIIb study TAMARA as an example and their comparison with traditional remission criteria. Ann Rheum Dis 2011;70:1986-90.
- 29 Korpela M, Laasonen L, Hannonen P et al. Retardation of joint damage in patients with early rheumatoid arthritis by initial aggressive treatment with disease-modifying antirheumatic drugs: five-year experience from the FIN-RACo study. Arthritis Rheum 2004;50:2072–81.
- 30 Montecucco C, Todoerti M, Sakellariou G, Scirè CA, Caporali R. Low-dose oral prednisone improves clinical and ultrasonographic remission rates in early rheumatoid arthritis: results of a 12-month open-label randomised study. Arthritis Res Ther 2012;14:R112.
- 31 Todoerti M, Scirè CA, Boffini N et al. Early disease control by low-dose prednisone comedication may affect the

- quality of remission in patients with early rheumatoid arthritis. Ann N Y Acad Sci 2010:1193:139–45.
- 32 Rantalaiho V, Korpela M, Hannonen P et al. The good initial response to therapy with a combination of traditional disease-modifying antirheumatic drugs is sustained over time: the eleven-year results of the Finnish rheumatoid arthritis combination therapy trial. Arthritis Rheum 2009; 60:1222-31.
- 33 van Everdingen AA, Siewertsz van Reesema DR, Jacobs JW, Bijlsma JW. The clinical effect of glucocorticoids in patients with rheumatoid arthritis may be masked by decreased use of additional therapies. Arthritis Rheum 2004;51:233–8.
- 34 Curtis JR, Westfall AO, Allison J *et al.* Population-based assessment of adverse events associated with long-term glucocorticoid use. Arthritis Rheum 2006;55:420-6.
- 35 Da Silva JA, Jacobs JW, Kirwan JR *et al.* Safety of low dose glucocorticoid treatment in rheumatoid arthritis: published evidence and prospective trial data. Ann Rheum Dis 2006:65:285–93.
- 36 Huscher D, Thiele K, Gromnica-Ihle E *et al.* Dose-related patterns of glucocorticoid-induced side effects. Ann Rheum Dis 2009;68:1119–24.
- 37 Borgna-Pignatti C, Rugolotto S, Nobili B et al. A trial of high-dose dexamethasone therapy for chronic idiopathic thrombocytopenic purpura in childhood. J Pediatr 1997; 130:13–16.
- 38 Fardet L, Flahault A, Kettaneh A et al. Corticosteroidinduced clinical adverse events: frequency, risk factors and patient's opinion. Br J Dermatol 2007;157:142–8.
- 39 Wassenberg S, Rau R, Steinfeld P, Zeidler H. Very low-dose prednisolone in early rheumatoid arthritis retards radiographic progression over two years: a multicenter, double-blind, placebo-controlled trial. Arthritis Rheum 2005;52:3371–80.
- 40 Klarenbeek NB, van der Kooij SM, Huizinga TJ et al. Blood pressure changes in patients with recent-onset rheumatoid arthritis treated with four different treatment strategies: a post hoc analysis from the BeSt trial. Ann Rheum Dis 2010;69:1342–5.
- 41 Hafström I, Rohani M, Deneberg S *et al.* Effects of low-dose prednisolone on endothelial function, atherosclerosis, and traditional risk factors for atherosclerosis in patients with rheumatoid arthritis—a randomized study. J Rheumatol 2007;34:1810–6.
- 42 Choy EH, Smith CM, Farewell V *et al*. Factorial randomised controlled trial of glucocorticoids and combination disease modifying drugs in early rheumatoid arthritis. Ann Rheum Dis 2008;67:656–63.
- 43 Buckley LM, Leib ES, Cartularo KS, Vacek PM, Cooper SM. Calcium and vitamin D3 supplementation prevents bone loss in the spine secondary to low-dose corticosteroids in patients with rheumatoid arthritis. A randomized, double-blind, placebo-controlled trial. Ann Intern Med 1996;125:961–8.
- 44 Buckley LM, Leib ES, Cartularo KS, Vacek PM, Cooper SM. Effects of low dose methotrexate on the bone mineral density of patients with rheumatoid arthritis. J Rheumatol 1997;24:1489–94.
- 45 Engvall IL, Svensson B, Tengstrand B, Brismar K, Hafström I. BARFOT study group. Impact of low-dose

1750

- prednisolone on bone synthesis and resorption in early rheumatoid arthritis: experiences from a two-year randomized study, 2008, http://arthritis-research.com/content/10/6/R128 (8 March 2014, date last accessed).
- 46 van Everdingen AA, Siewertsz van Reesema DR, Jacobs JW, Bijlsma JW. Low-dose glucocorticoids in early rheumatoid arthritis: discordant effects on bone mineral density and fractures? Clin Exp Rheumatol 2003;21: 155-60.
- 47 Coluccia D, Wolf OT, Kollias S et al. Glucocorticoid therapy-induced memory deficits: acute versus chronic effects. J Neurosci 2008;28:3474–8.
- 48 Grennan DM, Gray J, Loudon J, Fear S. Methotrexate and early postoperative complications in patients with rheumatoid arthritis undergoing elective orthopaedic surgery. Ann Rheum Dis 2001;60:214-7.
- 49 Kirwan JR, Bijlsma JW, Boers M, Shea BJ. Effects of glucocorticoids on radiological progression in rheumatoid arthritis. Cochrane Database Syst Rev 2007; (1):CD006356.
- 50 Gorter SL, Bijlsma JW, Cutolo M et al. Current evidence for the management of rheumatoid arthritis with glucocorticoids: a systematic literature review informing the EULAR recommendations for the management of rheumatoid arthritis. Ann Rheum Dis 2010;69:1010–4.
- 51 Keystone EC, Smolen J, van Riel P. Developing an effective treatment algorithm for rheumatoid arthritis. Rheumatology 2012l;51(Suppl 5):v48-54.
- 52 Jacobs JW. Lessons for the use of non-biologic anchor treatments for rheumatoid arthritis in the era of biologic therapies. Rheumatology 2012;51(Suppl 4):iv27–33.
- 53 Hoes JN, Jacobs JW, Buttgereit F, Bijlsma JW. Current view of glucocorticoid co-therapy with DMARDs in rheumatoid arthritis. Nat Rev Rheumatol 2010;6:693–702.
- 54 Bijlsma JW, van der Goes MC, Hoes JN et al. Low-dose glucocorticoid therapy in rheumatoid arthritis: an obligatory therapy. Ann N Y Acad Sci 2010;1193:123-6.
- 55 Joshi N, Rajeshwari K. Deflazacort. J Postgrad Med 2009;
- 56 Krogsgaard MR, Lund B, Johnsson B. A long-term prospective study of the equipotency between deflazacort and prednisolone in the treatment of patients with polymyalgia rheumatica. J Rheumatol 1995;22:1660-2.
- 57 Markham A, Bryson HM. Deflazacort. A review of its pharmacological properties and therapeutic efficacy. Drugs 1995;50:317–33.
- 58 Stahn C, Löwenberg M, Hommes DW, Buttgereit F. Molecular mechanisms of glucocorticoid action and selective glucocorticoid receptor agonists. Mol Cell Endocrinol 2007;275:71-8.
- 59 Cooper MS, Zhou H, Seibel MJ. Selective glucocorticoid receptor agonists: glucocorticoid therapy with no regrets? J Bone Miner Res 2012;27:2238-41.
- 60 Bodor N. Retrometabolic approaches for drug design and targeting. Pharmazie 1997;52:491-4.
- 61 Buris LF, Bodor N, Buris L. Loteprednol etabonate, a new soft steroid is effective in a rabbit acute experimental model for arthritis. Pharmazie 1999;54:58-61.
- 62 Quan LD, Purdue PE, Liu XM et al. Development of a macromolecular prodrug for the treatment of inflammatory

- arthritis: mechanisms involved in arthrotropism and sustained therapeutic efficacy. Arthritis Res Ther 2010;12: R170.
- 63 Yuan F, Nelson RK, Tabor DE *et al.* Dexamethasone prodrug treatment prevents nephritis in lupus-prone (NZB × NZW)F1 mice without causing systemic side effects. Arthritis Rheum 2012;64;4029–39.
- 64 Metselaar JM, Wauben MH, Wagenaar-Hilbers JP, Boerman OC, Storm G. Complete remission of experimental arthritis by joint targeting of glucocorticoids with long-circulating liposomes. Arthritis Rheum 2003;48: 2059-66.
- 65 Anderson R, Franch A, Castell M et al. Liposomal encapsulation enhances and prolongs the antiinflammatory effects of water-soluble dexamethasone phosphate in experimental adjuvant arthritis. Arthritis Res Ther 2010:12:R147.
- 66 Barrera P, Mulder S, Smetsers A et al. Long-circulating liposomal prednisolone versus pulse intramuscular methylprednisolone in patients with active rheumatoid arthritis [abstract]. Proceedings of the American College of Rheumatology (ACR) Scientific Meeting, 27 October 2008, San Francisco, Abstract L-10.
- 67 Buttgereit F, Doering G, Schaeffler A et al. Targeting pathophysiological rhythms: prednisone chronotherapy shows sustained efficacy in rheumatoid arthritis. Ann Rheum Dis 2010;69:1275–80.
- 68 Alten R. Chronotherapy with modified-release prednisone in patients with rheumatoid arthritis. Expert Rev Clin Immunol 2012;8:123–33.
- 69 Cutolo M, Villaggio B, Otsa K et al. Altered circadian rhythms in rheumatoid arthritis patients play a role in the disease's symptoms. Autoimmun Rev 2005;4: 497–502.
- 70 Myles AB, Schiller LF, Glass D, Daly JR. Single daily dose corticosteroid treatment. Ann Rheum Dis 1976;35:73-6.
- 71 Murthy MH, Rhymer AR, Wright V. Indomethacin or prednisolone at night in rheumatoid arthritis? Rheumatol Rehabil 1978;17:8–13.
- 72 Buttgereit F, Doering G, Schaeffler A et al. Efficacy of modified-release versus standard prednisone to reduce duration of morning stiffness of the joints in rheumatoid arthritis (CAPRA-1): a double-blind, randomised controlled trial. Lancet 2008;371:205-14.
- 73 Buttgereit F, Mehta D, Kirwan J et al. Low-dose prednisone chronotherapy for rheumatoid arthritis: a randomised clinical trial (CAPRA-2). Ann Rheum Dis 2013;72:204–10.
- 74 Alten R, Döring G, Cutolo M et al. Hypothalamus-pituitaryadrenal axis function in patients with rheumatoid arthritis treated with nighttime-release prednisone. J Rheumatol 2010;37:2025–31.
- 75 Silverman MN, Sternberg EM. Neuroendocrine-immune interactions in rheumatoid arthritis: mechanisms of glucocorticoid resistance. Neuroimmunomodulation 2008; 15:19-28.
- 76 Barnes PJ, Adcock IM. Glucocorticoid resistance in inflammatory diseases. Lancet 2009;373:1905–17.
- 77 Buttgereit F. How should impaired morning function in rheumatoid arthritis be treated? Scand J Rheumatol Suppl 2011;125:28–39.